Author’s response to reviews

Title: Using Pedometers to Increase Physical Activity in Overweight and Obese Women: A Pilot Study

Authors:

Sebely Pal (s.pal@curtin.edu.au)  
cheryl cheng (c.chengl@curtin.edu.au)  
Garry Egger (garryegger@ozemail.com.au)  
colin binns (cbinns@curtin.edu.au)  
robert donovan (r.donovan@curtin.edu.au)

Version: 2 Date: 16 April 2009

Author’s response to reviews:

clinical trial registration ow provided and included in abstract.